Author's response to reviews

Title: Indigenous well-being in four countries: An application of the UNDP'S Human Development Index to Indigenous Peoples in Australia, Canada, New Zealand, and the United States

Authors:

Martin Cooke (cooke@uwaterloo.ca)
Francis Mitrou (francism@ichr.uwa.edu.au)
David Lawrence (D.Lawrence@curtin.edu.au)
Eric Guimond (guimonde@ainc-inac.gc.ca)
Dan Beavon (beavond@ainc-inac.gc.ca)

Version: 3 Date: 18 October 2007

Author's response to reviews:

Editor
BMC International Health and Human Rights

Dear Editors,

Re: Indigenous Well-Being in Four Countries: An Application of the Human Development Index to Indigenous Peoples in Australia, Canada, New Zealand, and the United States

Thank you for accepting, in principle, our manuscript for publication.

No ethical approval was obtained for the study, as no individuals were involved in the research, and analysis was undertaken of secondary data sets in the public domain. A statement to this effect has now been included in the manuscript.

If you have any further questions about the paper, please do not hesitate to contact us.

Thank you again for considering our paper.

Yours faithfully,

Martin Cooke PhD
Assistant Professor,
Department of Health Studies and Gerontology
University of Waterloo
Waterloo, Ontario, Canada